Author's response to reviews

Title: Presentations of patients of poisoning and predictors of poisoning-related fatality: findings from a hospital-based prospective study

Authors:

Hsin-Ling Lee (hllee0913@gmail.com)
Hung-Jung Lin (cmh7900@mail.chimei.org.tw)
Ting-Yuan S Yeh (er5577@yahoo.com.tw)
Chih-Hsien Chi (chich@mail.ncku.edu.tw)
How-Ran Guo (hrguo@mail.ncku.edu.tw)

Version: 6 Date: 6 January 2008

Author's response to reviews:

Dear Sir,

Thank you very much for giving us the instructions to correct the format of the manuscript “Presentations of patients of poisoning and predictors of poisoning-related fatality: findings from a hospital-based prospective study” for publication in BMC Public Health. My colleagues and I are excited to have this privilege and made the following changes accordingly as:

Title: We changed the format to sentence case.

Author list: The full name of Dr. Yeh is Ting-Yuan Steve Yeh.

Footnote on the title page: We deleted the note “Part of the work described in this manuscript was presented at the meeting of the 16th Conference of the International Society for Environmental Epidemiology, New York City, New York, August 2004.” and removed the funding and support information from the title page.

Abstract: We updated the abstract in the submission system so that it is identical to that of the manuscript.

Figure title: We removed the title from the file and put the legend and title after the reference list.

We believe all the instructions have been followed. If you have any further requests or questions, please do not hesitate to contact me.

Sincerely yours,

How-Ran Guo, MD, MPH, ScD